Author’s response to reviews

Title: Screen-time Weight-loss Intervention Targeting Children at Home (SWITCH): A randomized controlled trial study protocol

Authors:

Ralph Maddison (r.maddison@ctru.auckland.ac.nz)
Cliona Ni Mhurchu (c.nimhurchu@ctru.auckland.ac.nz)
Louise Foley (l.foley@ctru.auckland.ac.nz)
Leonard Epstein (lhenet@buffalo.edu)
Yannan Jiang (y.jiang@ctru.auckland.ac.nz)
Midi Tsai (m.tsai@ctru.auckland.ac.nz)
Ofa Dewes (o.dewes@auckland.ac.nz)
Ihirangi Heke (ihi.heke@otago.ac.nz)

Version: 5  Date: 29 June 2011

Author’s response to reviews:

Abstract - last sentence changed to weight management programs
Page 6 - Outcomes are in italics - this needs to be changed
Page 11, final para. Removed word 'questionnaire” from Multimedia activity recall for children and adolescents
Page 13, last sentence - closed parentheses